Author’s response to reviews

Title: Expanding the spectrum of A20 haploinsufficiency in two Chinese families: cases report

Authors:
Guomin Li (liguomin486@sina.com)
Hai-mei Liu (liu__hm@163.com)
Wan-zhen Guan (gwzleo@163.com)
Hong Xu (hxu@shmu.edu.cn)
Bing-bing Wu (bingbingwu2010@163.com)
Li Sun (lillysun@263.net)

Version: 6 Date: 27 Jun 2019

Author’s response to reviews:

Dear editor:

Thank you for giving us a chance to revised our manuscript(MGTC-D-19-00024R5).

We read comments from editors carefully, and have revised our manuscript based on their comments.

Best wishes

Children’s Hospital of Fudan University

Guo-min Li
Editor Comments:

1- Please remove any files from the file inventory that you do not wish to see published.

We have deleted all files, which we do not wish to see published.

2- Any manuscript submitted to a BioMed Central journal must be original and the manuscript, or substantial parts of it, must not be under consideration by any other journal. We note that the current submission contains textual overlap with other previously published works. Please re-write these sentences and phrases in your own words to ensure no overlap. If there is overlap in the Methods section, please ensure that you summarize the methods and cite the source.

We have modified overlap with other publications in background, methods and discussion.

3- Please clarify whether consent to publish was obtained from the adult individual in addition to the parents of the child individuals. Please make this clear in your statement. Also, consent to publish is required from both the grandparents to be included in figure 1.

We have added statement in part of consent for publication. all people consent to publish.

4- At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Should you wish to respond to these revision requests, please include the information in the designated input box only.

This manuscript provide by us is a single, final, clean version.